Author's response to reviews

Title: Analysis of the effectiveness of interventions used during the 2009 A/H1N1 influenza pandemic

Authors:

Nilimesh Halder (nilimesh@csse.uwa.edu.au)
Joel K Kelso (joel@csse.uwa.edu.au)
George J Milne (milne@cse.uwa.edu.au)

Version: 6 Date: 23 March 2010

Author's response to reviews:

Authors' Contributions section updated to state that yes, all authors read and approved the final manuscript. Apologies for this omission.